ofgem Promoting choice and value for all gas and electricity customers

RIIO-T1 Electricity Transmission Price Control -Regulatory Instructions and Guidance: Version 1 DRAFT

Regulatory Instructions and Guidance

Reference:	139/12	Contact:	Martin Rodgers
Publication date:	30/10/12	Team:	RIIO-T1
		Tel:	0207 901 7273
		Email:	martin.rodgers@ofgem.gov.uk

Overview:

This document provides instructions and guidance to the three electricity transmission owners - National Grid Electricity Transmission plc, SP Transmission Ltd and Scottish Hydro Electric Transmission Ltd - to enable them to complete the reporting requirements associated with the transmission price control from 1 April 2013 to 31 March 2021.

This is the first transmission price control to reflect the new RIIO (Revenue = Incentives + Innovation + Outputs) model.

Context

This document contains the electricity transmission price control cost, outputs, financial and revenue Regulatory Instructions and Guidance (RIGs).

The purpose of this document is to provide a framework to allow Ofgem to collect accurate and consistent cost and revenue information from the three electricity transmission owners (TOs) - National Grid Electricity Transmission plc (NGET), SP Transmission Ltd (SPTL) and Scottish Hydro Electric Transmission Ltd (SHETL).

A number of licence conditions require the three electricity TOs to provide us with this information. The key licence condition for the purposes of this document is Standard Condition B15 [ETC74]: Regulatory Instructions and Guidance ('the RIGs Licence Condition').

This guidance applies for reporting from the period 1 April 2013 until 31 March 2021.

Associated documents

Supporting Documents

RIIO-T1 RIGs Electricity Transmission Reporting Template

RIGs Example Commentary

Other Documents

RIIO-T1: Initial Proposals for NGGT and NGET - Overview

RIIO-T1: Initial Proposals for NGGT and NGET - Outputs, incentives and innovation

RIIO-T1: Initial Proposals for NGET and NGGT – Cost assessment and uncertainty

RIIO-T1: Initial Proposals for NGGT and NGET – Finance

<u>RIIO-T1 and RIIO-GD1: Draft licence conditions – First informal licence drafting</u> <u>consultation</u>

<u>RIIO-T1: Initial Proposals for National Grid Electricity Transmission plc and National Grid</u> <u>Gas plc - Headlines</u>

Contents

1. Introduction	1
Background	1
Legal framework	1
Components of the RIGs	2
Overall structure	2
RIGs templates	2
Instructions and guidance	2 3
Commentaries	
Reporting under the RIGs	4 4
Timescales for reporting Resubmissions	4
Structure of this document	5
2. Instructions for completing the financial issues worksheets Introduction	6 6
Overview of worksheets	
Worksheet – Income statement	6 7 7
Worksheet – Financial position	7
Worksheet – Cash flow	8
Worksheet - Reconciliation of RRP costs to regulatory accounts	8
Worksheet – Net debt and interest	9
Worksheet – Net debt and interest 2 (additional NG table)	9
Worksheet – Net debt reconciliation	9
Worksheet – Fixed asset disposals	9
Worksheet – Tax computation	9
Worksheet –Tax allocations of spend	10
Worksheet –Tax allocations CT 600	10
Worksheet – Future financing requirements	11
Worksheet – Pensions Defined Benefit (DB) deficit	11
Worksheet – Pensions for benchmarking Worksheet – Pensions PPF and admin	12 14
Worksheet – Pension reconciliation triennial	14
Worksheet – Deficit Allocation Methodology (DAM) statement triennial	16
Worksheet – DAM reconciliation triennial	16
Worksheet – Pensions defined benefit triennial	16
3. Instructions for completing the total expenditure worksheets	20
Introduction	20
Overview of worksheets	20
Worksheet – Summary Totex (No input required)	20
Worksheet - Ofgem Adjustments to Totex (No input required)	20
4. Instructions for completing the operating expenditure workshe	ets
	21
Introduction	21
Overview of worksheets	21
Worksheet – Summary cash controllable costs	22
Worksheet – Provisions	23

Worksheet - Direct opex Worksheet - Analysis of excluded, consented, and de minimis services Worksheet - Analysis of CEO & other corporate costs Worksheet - Analysis of IS & telecoms costs Worksheet - IS & telecoms costs allocations Worksheet - Insurance costs Worksheet - Analysis of property costs by building Worksheet - Property costs allocations Worksheet - Property costs allocations Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience	23 24 24 24 25 25 25 25 25 26 26
 Worksheet - Analysis of CEO & other corporate costs Worksheet - Analysis of IS & telecoms costs Worksheet - IS & telecoms costs allocations Worksheet - Insurance costs Worksheet - Analysis of property costs by building Worksheet - Property costs allocations Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience 	24 24 24 25 25 25 25 25 25 25 25
Worksheet - Analysis of IS & telecoms costs Worksheet - IS & telecoms costs allocations Worksheet - Insurance costs Worksheet - Analysis of property costs by building Worksheet - Property costs allocations Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience	24 24 25 25 25 25 25 25 25 26
Worksheet - IS & telecoms costs allocations Worksheet - Insurance costs Worksheet - Analysis of property costs by building Worksheet - Property costs allocations Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience	24 25 25 25 25 25 25 25
Worksheet - Insurance costs Worksheet - Analysis of property costs by building Worksheet - Property costs allocations Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience	24 25 25 25 25 25 26
Worksheet - Analysis of property costs by building Worksheet - Property costs allocations Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience	25 25 25 25 25 25 26
Worksheet - Property costs allocations Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience	25 25 25 25 25 26
Worksheet - Analysis of business support costs Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet - Resilience	25 25 25 26
Worksheet - Business support allocations Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet – Resilience	25 25 26
Worksheet - Related party transactions Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet – Resilience	25 26
Worksheet - Year on year movement in controllable costs Worksheet - Total transmission salary and FTE numbers Worksheet – Resilience	26
Worksheet - Total transmission salary and FTE numbers Worksheet – Resilience	
Worksheet – Resilience	
	26
Worksheet – Apprentices and training	26
Worksheet – Insourced/Outsourced	27
Worksheet – Innovation Rollout Mechanism (IRM) expenditure	27
Worksheet – Network Innovation Allowance (NIA) expenditure	27
Worksheet – Network Innovation Competition (NIC) expenditure	27
Worksheet - Innovation revenue condition	28
Worksheet - Innovation revenue input	28
Worksheet - Physical security	28
5. Instructions for completing the capital expenditure worksheets	29
Introduction	29
Overview of worksheets	29
Worksheet – Capex summary	30
Worksheet – Non operational capex	30
Worksheet – System Operator (SO) capex	30
Worksheet - Load-related scheme listing	30
Worksheet - Load-related scheme annual profile	31
Worksheet - Non load-related scheme listing	31
Worksheet - Non load-related scheme annual profile	31
Worksheet - Other capex	31
Worksheet - Flood mitigation	31
Worksheet – Transmission Investment Renewable Generation (TIRG) /	
Transmission Investment Incentives (TII)/ Strategic Wider Works (SWW)	
schemes	32
Worksheet – Unit costs actual	32
	32
Worksheet – Unit costs actual	
Worksheet – Unit costs actual Worksheet - Unit costs future levels	32
Worksheet – Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex	32 32
Worksheet – Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets	32 32 33
Worksheet – Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets Introduction	32 32 33 33 33 34
Worksheet – Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets Introduction Overview of worksheets Worksheet - System characteristics Worksheet – Activity indicators	32 32 33 33 33 34 34
Worksheet – Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets Introduction Overview of worksheets Worksheet - System characteristics Worksheet - Activity indicators Worksheet - Activity costs	32 32 33 33 33 34 34 34
Worksheet - Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets Introduction Overview of worksheets Worksheet - System characteristics Worksheet - System characteristics Worksheet - Activity indicators Worksheet - Activity costs Worksheet - Transmission system performance	32 32 33 33 34 34 34 34 34
Worksheet - Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets Introduction Overview of worksheets Worksheet - System characteristics Worksheet - System characteristics Worksheet - Activity indicators Worksheet - Activity costs Worksheet - Transmission system performance Worksheet - Faults	32 32 33 33 33 34 34 34 34 34 34
Worksheet - Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets Introduction Overview of worksheets Worksheet - System characteristics Worksheet - System characteristics Worksheet - Activity indicators Worksheet - Activity costs Worksheet - Transmission system performance Worksheet - Faults Worksheet - Faults Worksheet - Failures	32 32 33 33 34 34 34 34 34 34 35
Worksheet - Unit costs actual Worksheet - Unit costs future levels Worksheet - Physical security capex 6. Instructions for electricity network data worksheets Introduction Overview of worksheets Worksheet - System characteristics Worksheet - System characteristics Worksheet - Activity indicators Worksheet - Activity costs Worksheet - Transmission system performance Worksheet - Faults	32 32 33 33 33 34 34 34 34 34 34

Worksheet – Demand and supply at substations	36
Worksheet – Asset description	36
Worksheet – Asset age profile	36
Worksheet – Asset disposals load-related	36
Worksheet – Asset disposals non load-related	36
Worksheet - Asset additions and disposals total (Aged Based Modelling)	37
Worksheet – Asset additions and disposals total	37
Worksheet – Asset replacement priority driven disposals	37
Worksheet – Asset lives	37
7. Instructions for completing the outputs worksheets	38
Introduction	38
Overview of worksheets	38
Worksheet – Safety	39
Worksheet – Reliability	40
Worksheet – Customer satisfaction	40
Worksheet – Timely connections	41
Worksheet – Environment SF ₆ emissions	41
Worksheet – Visual amenity designated areas	41
Worksheet – Strategic Wider Works (SWW) outputs	42
Worksheet – Pre-construction deliverables	42
Worksheet – Environment Business Carbon Footprint	42
Worksheet – NGET wider works volume driver	47
Worksheet – NGET local generation and demand volume driver - summary	47
Worksheet – NGET local generation volume driver – scheme level detail	48
Worksheet – NGET planning requirements volume drivers (Uncertainty	40
Mechanism)	48
Worksheet – NGET DNO mitigation schemes - scheme level detail	49
Worksheet – NGET undergrounding volume driver – scheme level detail	49
Worksheet – SHETL local generation volume driver	50 51
Worksheet – Generation connections SPTL Worksheet – Collector substations volume driver	51
	51
Worksheet - Network Output Measures (NOMS) Worksheet – NOMS NGET	51
Worksheet – Criticality substations	51
Worksheet – Criticality substations Worksheet – Criticality circuits	51
Worksheet – Criticality SPTL	52
8. Instructions for completing the revenue reporting worksheets Introduction	53
Appendices	54
Appendix 1 – Glossary and Definitions	

1. Introduction

Chapter Summary

This chapter sets out the purpose and structure of the Regulatory Instructions and Guidance (RIGs) which will apply for the electricity transmission owners for RIIO-T1. It also sets out some guidance on the process for reporting under the RIGs and our audit requirements.

Background

1.1. This is the first price control to be conducted under our new RIIO (Revenue = Incentives + Innovation + Outputs) model. Through RIIO-T1, we are setting the regulatory framework to apply to electricity and gas transmission companies from 1 April 2013 to 31 March 2021.

1.2. The Regulatory Instructions and Guidance (RIGs) provide a framework which enables Ofgem to collect data from the electricity transmission owners (TOs) during the RIIO-T1 period. We collect data to enable us to administer the Special Licence Conditions of the TOs (the conditions which relate to the price control) and our Final Price Control Proposals. For example, the RIGs allow us to monitor TOs performance against the outputs that they are required to deliver, to calculate any rewards or penalties associated with incentive mechanisms, and to determine additional cost allowances determined within period, ie costs determined through uncertainty mechanisms. The RIGs also provide a database of TO performance which we draw on to set cost allowances at subsequent review periods.

1.3. The RIGs enable Ofgem to collect data from TOs in a consistent format. The RIGs inform the TOs about the information we plan to collect, guide them on how to provide this information and enable the TOs to put the systems in place to collect the data to the detail we require.

Legal framework

1.4. For RIIO-T1 the reporting requirements have been consolidated in a single new licence condition:

• Standard Condition B15 [ETC74]: Regulatory Instructions and Guidance ('the RIGs Licence Condition').

1.5. The new licence condition sets out the scope, content and common governance arrangements for the RIGs.

1.6. The purpose of introducing the proposed new RIGs Licence Condition is to consolidate the reporting related provisions, which were previously included within a number of licence conditions into a single condition. Having a single licence condition

has not altered the broad types of information that the energy network companies will need to provide.

Components of the RIGs

Overall structure

- 1.7. The RIGs comprise four main elements:
 - templates for reporting data (in MS Excel v.2007)
 - templates containing the calculation of the revenue elements revenue reporting (including performance against incentives) (in MS Excel v.2007)
 - templates for providing commentaries against the data (in MS Word)
 - instructions and guidance on how to report the data (in MS Word).

RIGs templates

1.8. The data templates have been designed using the RIIO-T1 business plan data templates alongside the previous price control RIGs and Regulatory Reporting Packs (RRPs). Added to these are the new data requirements arising from the new outputs, incentives and obligations from RIIO-T1.

1.9. The key points to note in completing the RIGs templates are:

- Some of the workbooks in these RIGs have been designed to link to other workbooks. These links must be retained by the TO in the version submitted to Ofgem. Failure to do so will be considered non-compliant with the RIGs.
- Financial values should be provide in £ million figures and displayed to one decimal place. However, the TOs are required to provide data in the RIGs to the highest level of accuracy available to them with the minimum being three decimal places, unless otherwise indicated in the guidance.
- Financial values should be input into the tables (yellow cells) as positive values unless instructed otherwise.
- Any restatement of historical values should be clearly highlighted and explained.

Instructions and guidance

1.10. To enable the TOs to complete the associated workbooks we provide supporting instructions and guidance. This document constitutes Version 1 of the instructions and guidance for the TOs for RIIO-T1.

1.11. The purpose of this document is to provide instructions and guidance to enable the TOs to complete the associated workbooks. This document provides information on:

- the systems, processes, procedures, recording and provision of the required data
- reporting units and levels of accuracy (including rounding)
- the methodology for calculating or deriving required numbers
- the provision of the data to the Authority (format, frequency etc)
- any audit or examiner requirements
- reasons why the data is required
- explanations of how the Authority will monitor, assess, and enforce compliance
- a glossary of terms used in the workbooks.

1.12. We set out further information on this document below under the heading 'Structure of this document'.

Commentaries

1.13. Alongside the submission of its templates, each TO must complete the associated commentary. There are three main drivers for the completion of the commentary templates:

- (1) Where data is not captured on the same basis as required by the reporting definitions. Where a TO does not capture data on the same basis as the reporting definitions, and in completing the tables the licensee has to apportion costs across one or more activities, the basis of apportionment must be provided in the cost commentary document. The licensee must also give an indication of their assessment of the robustness of those assumptions. In such instances, the licensee should explain the reasons why it has been unable to capture the data on the same basis as the cost reporting definitions and identify any steps that it will take to ensure that the data is captured in accordance with the relevant definitions in future years.
- (2) Where a modification of the RIGs changes the reporting categories or activities. In such cases the TO must provide estimates in respect of that category or activity, for the year in which the modification is made and for any preceding year, derived from other information available to the licensee as may be appropriate for that purpose. In this case the licensee must set out the basis and methodology for deriving the estimated amounts in detail in the commentary.
- (3) **To explain why costs have been incurred and the annual movements in cost levels.** This explanation will be used in conjunction with the data tables, to understand the structures and operations of each TO, to monitor each TO's performance against its business plan and funding included in the RIIO-T1 Final Proposals document.

Reporting under the RIGs

Timescales for reporting

1.14. The relevant reporting year for the provision of information under the RIGs is from 1 April to 31 March the following calendar year.

1.15. Except where otherwise stated, the TOs must provide the information required under the RIGs as soon as reasonably practicable and in any event not later than 31 July following the end of the relevant reporting year to which such information relates. This is the latest date that TOs can submit information unless the Authority has previously consented to an extension in writing.

Form of submission

1.16. Instructions for the electronic submission of the workbooks will be circulated to each licensee's regulation managers in advance of each submission deadline. However, if there is any doubt about the method of submission, the licensee must contact Ofgem for clarification.

1.17. The submission must be accompanied by a letter signed by a director on behalf of the licensee confirming that the data is accurate and has been provided in accordance with the RIGs.

Resubmissions

1.18. The TOs are required to seek the agreement of the Authority before resubmitting any information provided in accordance with these RIGs.

1.19. The volume of information the company is required to provide will depend on the nature of the required resubmission. Where significant changes have been made the resubmission should be accompanied by a letter signed by a director.

1.20. For each resubmission a separate explanation must be provided listing every cell that has been amended. The explanation must include sufficient commentary to explain the reasons for the required resubmission.

1.21. In relation to the detailed return required as part of revenue reporting, this must only be resubmitted where a restatement is necessary in the opinion of the appropriate auditor.

Review

1.22. Once the TOs have submitted the information to the Authority, Ofgem or a person nominated by the Authority ('a reviewer') will undertake a detailed review of the information. A review may include a visit to each TO for discussion of the information submitted. Such visits will be agreed with the licensees in advance.

1.23. Where a reviewer has been nominated, the reviewer will enter into an agreement with the licensee to maintain confidentiality on reasonable terms.

Appointing an examiner

1.24. In accordance with the RIGs Licence Condition the licensee must permit a person nominated by the Authority to examine:

- the systems, processes and procedures for measuring the specified information
- the specified information collected by the licensee
- the extent to which the systems, process and procedures and the specified information complies with the RIGs.

Audit requirements in relation to revenue reporting

1.25. In accordance with the RIGs Licence Condition, Ofgem will identify the Specified Information which is to be subject to audit, the terms on which an auditor is to be appointed by the licensee for that purpose and the nature of the audit to be carried out by that person. We will issue an Agreed Upon Audit Procedures (AUP) for use by an appropriate auditor.

Structure of this document

1.26. This document is divided into sections reflecting the different component parts of the RIGs workbooks. These are as follows:

- Chapter 2 provides instructions for the completion of the financial issues worksheets.
- Chapter 3 provides instructions for the completion of the total expenditure worksheets.
- Chapter 4 provides instructions for the completion of the operating expenditure worksheets.
- Chapter 5 provides instructions for the completion of the capital expenditure worksheets.
- Chapter 6 provides instructions for the completion of the electricity network data worksheets.
- Chapter 7 provides instructions for the completion of the outputs worksheets.
- Chapter 8 provides instructions for the completion of the revenue reporting worksheets.

1.27. Appendix 1 sets out a glossary and definitions of the terms included in these instructions and in the associated worksheets.

2. Instructions for completing the financial issues worksheets

Chapter Summary

The purpose of this chapter is to inform the completion of the financial issues worksheets by each TO. This is to enable Ofgem to effectively monitor outturn against the appropriate allowances, assess future allowances and quantify any adjustments as required in the Final Proposals.

Introduction

2.1. The purpose of these worksheets is to facilitate the submission of uniform and comparable financial information from licensees. This enables comparison with business plans, comparison with prior years and comparative regulation on a consistent basis throughout the RIIO-T1 period.

2.2. Licensees should submit accurate (and where instructed) audited figures of their costs and revenues for the relevant period. Further guidance is provided below.

2.3. All costs are to be entered on a Cash Basis (see Glossary) and exclusive of atypical items except where specifically instructed to report data. Cash means exclusive of all provisions and all accruals and prepayments that are not incurred as part of the ordinary level of business.

Overview of worksheets

2.4. The worksheets included within this chapter are:

- Income statement
- Financial position
- Cash flow
- Reconciliation to regulatory accounts
- Net debt schedule
- Net debt NG
- Net debt reconciliation National Grid
- Fixed asset disposals
- Tax computation
- Tax pools
- Tax allocations of spend

- Tax allocations CT 600
- Financing requirements
- Pensions defined benefit deficit
- Pensions for benchmark
- Pensions PPF and admin
- Pension reconciliation triennial
- Deficit allocation methodology (DAM) statement triennial
- DAM reconciliation triennial
- Pensions defined benefit triennial

Worksheet – Income statement

2.5. The purpose of this table is to obtain a standard form of each licensee's Balance Sheet / Statement of financial position, consistent with the Regulatory accounts. It will be used to provide the data in this statement in the regulatory accounts on a consistent and comparable basis across Licensees and time; and to populate the financial model. It will be used to check the level of total debt reported elsewhere and what other liabilities the licensee has; and to populate the financial model. It will also be used to identify differences in reporting when Licensees switch from UK GAAP to IFRS during the price control period.

2.6. This table should be populated from and agree to the entity's audited Regulatory accounts for the relevant year. Whilst there are more categories than in the Profit & Loss / Statement of comprehensive income in audited regulatory accounts the additional detail is required to facilitate reconciliation/agreement with other tables. Specifically:

- Row 9: Indicate whether accounts are prepared under UK GAAP or IFRS accounting standards.
- Row 13: Insert total revenue as shown in the regulatory accounts.
- Rows 13 to 59: These rows cover the licensed entity and should agree to the audited Regulatory accounts.
- Row 37: "Other finance income/Investment income" on the P&L account should include non-transportation related items, financial movements on pension schemes and all movements on derivative financial instruments.

Worksheet – Financial position

2.7. The purpose of this table is to obtain a standard form of each licensee's Balance Sheet / Statement of financial position, consistent with the Regulatory accounts. It will be used to provide the data in this statement in the regulatory accounts on a consistent and comparable basis across Licensees and time; and to populate the financial model. It will be used to check the level of total debt reported elsewhere and what other liabilities the licensee has; and to populate the financial model. It will also be used to identify differences in reporting when Licensees switch from UK GAAP to IFRS during the price control period.

2.8. This table should be populated from, and agree to, the audited Regulatory accounts for the relevant year. Whilst there are more categories than in the Balance Sheet / Statement of financial position in the entity's audited regulatory accounts the additional detail is required to facilitate reconciliation/agreement with other tables.

- Row 62: The "Hedging Reserve" on the balance sheet is only to be used by companies reporting under IFRS.
- Row 70: The net debt shown is calculated based on the definition of net debt issued in the open letter of 31 July 2009 "Clawback of tax benefit due to excess gearing".
- Rows 268 to 317: Where the entity has reported under IFRS for the first time, this section should be completed to identify the differences between the two approaches.
- Rows 321 to 369: Enter details of the book values of the disposal of tangible and intangible assets. Enter the corresponding cash proceeds split for tangible assets into operational and non-operational assets. The cash proceeds should be entered as a negative value, except where the disposal is to an affiliate where market values should be entered again as a negative value. Disposal information not required for 2012/13 as covered in Table 1.7.

Worksheet – Cash flow

2.9. The purpose of this table is to obtain a standard form of each licensee's cash flow position consistent with the Regulatory Accounts. It will be used to provide the data in this statement in the regulatory accounts on a consistent and comparable basis across Licensees and time; and to sense check the financial model.

2.10. This sheet is linked to movements implied by the Profit and Loss (P&L) and balance sheet as far as practicable. Inputs are required to replicate the cash flow in the regulatory accounts.

• Rows 48 to 55: Reconciliation is required between opening debt and the closing debt implied by the decrease/ increase in net cash.

Worksheet - Reconciliation of RRP costs to regulatory accounts

2.11. The purpose of this table is to allow the licensee to provide details and numerical explanation of variances between Regulatory Accounts and the prime accounting statements in the RRP.

2.12. Input narrative describing what line and which section of the Regulatory Accounts or the RRP the particular amount of the reconciliation relates is required. A few words of explanation of the reason for the variance should be provided either in the RRP accompanying narrative or on the worksheet below the reconciliation table. Input the relevant figures to reconcile the differing amounts.

Worksheet – Net debt and interest

2.13. The purpose of this table is to collect details of all borrowing, guarantees and financial derivatives, interest rates on debt, repayment dates and cash and short term investments. This information is used to compare gearing levels to the regulatory modelling data; and to enable the calculation of the tax benefit of excess gearing. It is also used to provide details of future refinancing requirements where DNOs have loans falling due for repayment. The table provides information on derivative financial instruments to facilitate understanding of existing debt structures. The table is designed to enable different types of borrowing and lending to be identified and any derivative products relating to the debt to be linked to it.

2.14. There are no specific instructions for completion of the table.

Worksheet – Net debt and interest 2 (additional NG table)

2.15. The purpose of this table is for National Grid to provide additional analysis of their Net Debt information (as detailed in the sheet).

2.16. There are no specific instructions for completion of the table.

Worksheet – Net debt reconciliation

2.17. The purpose of this table is for licensees to provide details and reasons for any difference between Net Debt as stated in the Regulatory Accounts and the RRP.

2.18. There are no specific instructions for completion of the table.

Worksheet – Fixed asset disposals

2.19. The purpose of this table is to collect information relating to fixed asset disposals to assist the reconciliation of the financial model to accounts.

2.20. The table should detail assets transferred from the Transmission business to a company within the same group ie a property company.

Worksheet – Tax computation

2.21. This worksheet provides an analysis of taxation. It will enable the Authority to confirm the split by business and form of control to compare against allowances.

2.22. There are no specific instructions for completion of the table.

Worksheet – Tax pools

2.23. This table must be prepared on the same basis as the licensee's corporation tax returns for each regulatory financial year ended 31 March and in accordance with extant tax legislation [Note this has yet to be done – only shown for licensee as a whole not form of control]

2.24. This table is used to track the annual additions in the regulatory financial year and movements in capital allowances for the licensee, and by the individual business segments.

2.25. It will capture the movements on capital allowances in total for the licensee and for the distribution business and other activities; and reconcile the total to the licensee's own tax provision workings for its regulatory accounts and/or draft corporation tax computation.

2.26. It is also used to identify prior year adjustments arising in the individual capital allowance pools from the closing and settlement of open tax years.

2.27. The closing pool balances will be used to populate the financial model at the next price control, subject to review

Worksheet –Tax allocations of spend

2.28. The purpose of this table is to report the allocation of capex and opex spend to capital allowance pools for the licensee consistent with the numbers in the Regulatory Accounts for the year. It obtains an analysis of the TO's tangible fixed asset additions and ensures allocation to the capital allowance pools on a consistent and comparable format across Licensees. It enables monitoring and comparison of allocations to capital allowance pools across Licensees, and across time, which will inform future price controls.

2.29. The table requires Licensees to insert the allocation of expenditure across tax categories for the various types of expenditure. These allocations should reflect the basis on which the tax charge for the Regulatory Accounts has been based. Enter the analysis of the expenditure in each area for tax purposes (\pounds m) by each of the capital allowance pools. The total expenditure should agree to that reflected in the capex summary table. The allocation of indirect costs between capitalised and non capitalised should be entered in the section titled 'Recharges to capex of indirect business costs'. Profiles are calculated for each expenditure type in the bottom half of the sheet.

Worksheet – Tax allocations CT 600

2.30. The purpose of this table is to report the allocation of capex and opex spend to tax pools for the licensee consistent with the CT600 tax return submitted (normally for the preceding statutory accounting financial reporting year). It obtains an analysis of the NWO's tangible fixed asset additions and ensures allocation to the capital allowance pools on a consistent and comparable format across Licensees. It

enables monitoring and comparison of allocations to capital allowance pools across Licensees, and across time, which will inform future price controls.

2.31. Enter the year end to which the CT600 refers in the closest column to the regulatory year end (eg a 31/12/08 CT600 return would be entered in the column for regulatory year end 31/03/09). The table requires licensees to insert the allocation of expenditure across tax categories for the various types of expenditure. The completion is consistent with the guidance for Table 1.4.1 (at a more detailed level) although there is no check to the capex summary since the two are not necessarily aligned.

2.32. The total additions by capital allowance pool should agree to the additions stated in the CT600. The final section provides a segmental analysis of the capital additions. Entries should be made in each appropriate section where pool additions do not relate to the transmission business.

Worksheet – Future financing requirements

2.33. The purpose of this table is to inform consideration of the licensee's short term financial stability. It requires provision of a high-level cashflow for the succeeding two regulatory years and the financing requirements over the same period.

2.34. Inputs the figures indicated into the yellow shaded boxes. The figures required are for the next two full regulatory financial years. Row 29: This line requires completion for the value available at the start of the current year. Insert the value of any undrawn facilities which will be available to meet the financing requirements. Where the licensee has concerns about refinancing, these should be expressed in the commentary, except to the extent that such issue have been separately notified to Ofgem.

Worksheet – Pensions Defined Benefit (DB) deficit

2.35. The purpose of this table is to collect recent data relating to defined benefit scheme, obtained from pension scheme accounts and updated valuations. This information provides a view of scheme investment and funding strategies in between triennial true up and resets. The table collects information on actual pension contributions made by the licensee to repair the deficit; estimated movements in assets, liabilities and membership data.

2.36. All data should be entered in nominal prices each year. Data is to be provided on the total DB scheme for which the licensee is a sponsoring employer. This section collects information on the total DB scheme for which the licensee is a sponsoring employer.

2.37. Section A membership data Rows 7 – 13: This section seeks information on the membership of the total DB scheme. It is accepted that this data may need to be revised in subsequent years.

2.38. Section B scheme assets:

- Rows 18 40: Enter details on the movements in assets held in the scheme. The balance brought forward should be the gross value of assets in the scheme. If opening balances are revised, please correct the previous year"s data rather than adjusting in the current year. Where any changes are made to prior year data, these should be detailed in the Commentary. Enter any contributions into the scheme as positive numbers. Benefit payments made by the scheme and other withdrawals should be shown as a negative. Investment profits should be entered as a positive, whereas investment losses should be entered as a negative.
- Rows 47 71: Enter detail of assets held by the scheme. Where other assets are identified, enter these in rows 65 to 71 and provide description in the commentary.

2.39. Section C scheme liabilities Rows 76 – 86: Enter details of the movements in pension scheme liabilities by category. Additional categories should be entered in rows 83 to 85.

2.40. Section D scheme deficit:

- Rows 90 107: Actual deficit repair payments and funding of pension related severance costs should be entered into row 92 and 94, respectively.
- Rows 97 107, total actual deficit repair payments (reported in row 92) should be attributed to the businesses within the sponsoring employer.
- Rows 100 105, please provide the names of all other licensees making deficit payments to the scheme; along with the amount of payments attributable.
- Rows 110 120: Input any actual payments made by the sponsoring employer towards a contingent asset (eg letter of credit or escrow account) for each business within the sponsoring employer.
- Rows 113 118, please provide the names of all other licensees making deficit payments to the scheme; along with the amount of payments attributable.

Worksheet – Pensions for benchmarking

2.41. The purpose of this table is to collect data on ongoing pension costs of licensees' DB scheme and defined contribution scheme(s), personal accounts and stakeholder schemes (if any). These form part of labour costs that will be included in the benchmarking analysis.

2.42. The table collects information about the licensee's pension costs rather than the total Defined Benefit and Defined Contribution (DC) schemes as a whole.

2.43. Section A – DB Scheme(s) ongoing contributions. Data is to be provided for each DB pension scheme where there is an element relating to the distribution business.

2.44. Pensionable pay data - DB scheme:

- Row 9: Enter total pensionable pay for the licensee scheme only. This should be before deduction of salary sacrifice arrangements. This figure will not necessarily be the same as Labour costs in the Cost tables.
- Row 10: The salary sacrifice element relates solely to sacrifice for additional DB pension contributions made under an arrangement with the sponsoring employer.

2.45. Employee contribution data – DB scheme:

- Row 13: Input employee contributions to the DB scheme as a percentage of pensionable pay.
- Row 14: Enter employees' cash contributions (£m) for ongoing costs to the DB scheme.
- Rows 18 21: Input employer contributions to the DB scheme as a percentage of pensionable pay. In so far as pension scheme administration costs and PPF levy are included in normal contributions for this scheme, these should be included in the percentage of pensionable pay calculation.
- Row 23: Enter the employer's cash contribution (£m) for ongoing costs to the DB scheme.

2.46. Attribution of licensee ongoing DB pension costs Rows 27 to 33: Enter the attribution of DB ongoing costs across the segments shown.

2.47. Section B – DC Scheme(s) contributions: Information is to be reported at the licensee level only.

2.48. Pensionable pay data - DC scheme: Row 38: Enter total pensionable pay for the licensee scheme only. This should be before deduction of salary sacrifice arrangements. This figure will not necessarily be the same as Labour costs in the Cost tables.

2.49. Contribution data – DC scheme:

• Row 40: Enter the cash contribution for the primary DC scheme.

- Rows 40 42: Enter the cash contribution to all stakeholder schemes and personal accounts.
- Row 45: Input employee contributions to the DC scheme as a percentage of pensionable pay.
- Row 46: Input sponsoring employer contributions to the DC scheme as a percentage of pensionable pay.

2.50. Attribution of licensee ongoing DB pension costs Rows 50 to 56: Enter the attribution of the defined contribution/ stakeholder costs across the segments shown.

2.51. Section C – Incremental Deficit Summary: This section will be developed as part of the Pension Deficit Allocation Methodology.

Worksheet – Pensions PPF and admin

2.52. The purpose of this table is to collect data, on a cash basis, on the Pensions Protection Fund (PPF) levies and pension administrative costs for each scheme. It shows the allocation of these costs between sponsoring entities. For the avoidance of doubt, pension scheme administration costs are the costs incurred by trustees in the running of the pension scheme and excludes investment management fees for each scheme. These do not include costs incurred by the licensee or other sponsoring employers in collecting employee contributions or managing their interface with the scheme. This enables reconciliation to the cost tables and allocation to totex where appropriate.

2.53. Data is to be provided on PPF levies and admin costs relating to all DB schemes for which the licensee is a sponsoring employer.

2.54. Section A – Total PPF levies paid by all DB schemes:

- Row 7: Enter the total cash payment in year for the PPF levy covering all DB schemes (including the levy for PPF administration charge). This may be paid either directly by the sponsoring employer or through the sponsoring employer's normal ongoing pension contributions.
- Rows 10 19: Enter the allocation of PPF levies paid in year direct by the employer (not collected through the scheme contributions), split by non-regulated and regulated business segments.
- Rows 13 18: Please provide the names of all other licensees making direct payments to the PPF levy; along with the amount of payments attributable.
- Rows 22 31: Enter the allocation of PPF levies, collected during the year through the scheme contributions (not paid direct by the employer), split by non-regulated and regulated business elements.

- Rows 25 30: Please provide the names of all other licensees making payments for PPF levies through normal ongoing contributions; along with the amount of payments attributable.
- Row 34: This row automatically calculates the effective PPF levy contribution as a percentage of pensionable pay.
- Rows 38 42: Enter the values of PPF levy that are attributed to the segments of the transportation businesses.
- Rows 46 47: Provide latest Dun & Bradstreet PPF failure score, which will be used by the PPF to calculate the risk-based levy; along with the date it was scored for the DB pension scheme. If the score is based on multi-company data, enter the weighted average score (as used by the PPF) and detail the constituent scores in the Commentary.

2.55. Section B – Pension admin costs paid by all DB schemes:

- Row 52: Enter the total pension administration costs for all DB schemes for which the licensee is a sponsoring employer.
- Rows 55 64: Enter the allocation of DB scheme administration costs paid in year directly by the employer (not collected through the scheme contributions), split by non-regulated and regulated business segments.
- Rows 58 63: Please provide the names of all other licensees making direct payments towards DB scheme administration costs; along with the amount of payments attributable.
- Rows 67 76: Enter the allocation of DB scheme administration costs, collected during the year through the scheme contributions (not paid directly by the employer), split by non-regulated and regulated business elements.
- Rows 70 75: Please provide the names of all other licensees making payments towards DB scheme administration costs through normal ongoing contributions; along with the amount of payments attributable.
- Rows 83 88: Enter the values of administration costs that are attributed to the segments of the transportation business.

Worksheet – Pension reconciliation triennial

2.56. The purpose of the table is to reconcile the value of pension costs shown in the cost reporting tables with the pension costs shown in the finance tables. It operates as a consistency check across the tables and also shows the total pension costs for comparison to price control allowances.

2.57. Instructions for completion are as follows:

- Rows 7 16 these rows automatically collect data from RIGs cost tables.
- Rows 19 25: Input any reconciling items between the total in the cost tables and the pension table totals (row 31).
- Rows 28 36: These rows automatically collect data from RIGs pension tables 1.9.1, 1.9.2 and 1.9.3.
- Rows 41 to 54: Insert any reconciling items between total pension costs as reported in the cost tables (row 11) and to that reported in the P&L and balance sheet movement figures in the Regulatory Accounts.
- Rows 59 to 72: Insert any reconciling items that have been reported in the accounts of related parties.
- Row 74: This row highlights any remaining reconciling balance, ie where the reported amounts in the Cost RIGs (row 11) does not reconcile to the amounts reported in the Financial Issues RIGs (row 38). The necessary tables must be reworked until the reconciling balance in row 74 is nil.
- Row 78: Enter pension charges as reported in the regulatory accounts P&L account statement.
- Rows 78 79: Insert any movement in pension provisions as reported in the regulatory accounts balance sheet statement.

Worksheet – Deficit Allocation Methodology (DAM) statement triennial

2.58. This table is to be defined as part of developing the Pension Deficit Allocation Methodology (DAM).

Worksheet – DAM reconciliation triennial

2.59. This table is to be defined as part of developing the DAM.

Worksheet – Pensions defined benefit triennial

2.60. The purpose of this table is to collect data relating to defined benefit scheme pension costs, assets and liabilities, deficits and ancillary data. The table also collects information on all membership data and scheme data. This information will form a part of the panel data for the pension reasonableness reviews in accordance with our approach to pension costs and as set out in our Pension Principles.

2.61. Data is to be provided for each DB pension scheme where there is an element relating to the transmission business. During RIIO-T1, this table should be completed for columns G, J and M; with information at as 31 March 2013, 2016 and 2019, respectively.

2.62. Section A: This section collects information on the total DB scheme for which the licensee is a sponsoring employer. This data should be that for the latest updated s.224 valuation or latest full triennial valuation. It should not be information repeated from the previous triennial valuation. Instructions for completion are as follows:

- Rows 10 to 18: Enter the attribution of total scheme deficit or surplus by the constituent sponsoring affiliates, ie the total for Non-regulated businesses and the amounts attributable to each affiliated licensee. The method of attribution is to be detailed in the Commentary. Where any deficit is allocated to other licence holders, the attribution should be reflected in that licensee's RRP.
- Row 21: State the basis of the scheme valuation used (eg full triennial, Pension Act 2004 section 224 update).
- Row 26: Specify whether the scheme valuation has been audited.
- Rows 28 to 52: Enter details on the movements in assets held in the scheme. The balance brought forward should be the gross value of assets in the scheme. If opening balances are revised, please correct the previous year's data rather than adjusting in the current year. Where any changes are made to prior year data, these should be detailed in the Commentary. Sign Convention: Enter any contributions into the scheme as positive numbers. Benefit payments made by the scheme and other withdrawals should be shown as a negative. Investment profits should be entered as a positive, whereas investment losses should be entered as a negative.
- Rows 57 to 86: Enter detail of assets held by the scheme. Where other assets are identified, enter these in rows 78 to 85 and provide description in the commentary.
- Rows 91 to 105: Enter details of the movements in pension scheme liabilities by category. Additional categories should be entered in rows 100 to 105.
- Rows 111 to 128: Enter the details of all bulk transfers from corporate transactions since the previous reasonable review of pensions.
- Rows 111 116: Insert the names of any pension scheme that are counterparties to bulk transfers in or out of the licensee's scheme. Enter the year end value of the assets attributable from each 'bulk transfer' arising from a corporate transaction. Provide details in the Commentary, including how the valuations of assets and liabilities have been determined, eg actuarial assessment of assets and liabilities of relevant members. On the basis that the assets of a scheme may not be hypothecated or identifiable to any sub

group of members, and then the assets may be apportioned based on the ratio that the liabilities of these members bear to the total liabilities. The liabilities should be separately computed based on each member transferred in.

 Rows 120 – 125: Insert the names of any pension scheme that are counterparties to bulk transfers in or out of the licensee's scheme. Enter the year end value of the liabilities from each bulk transfers arising from a corporate transaction. Provide details in the Commentary, including how the valuations of assets and liabilities have been determined, eg actuarial assessment of assets and liabilities of relevant members. On the basis that the assets of a scheme may not be hypothecated or identifiable to any sub group of members, then the assets may be apportioned based on the ratio that the liabilities of these members bear to the total liabilities. The liabilities should be separately computed based on each member transferred in.

2.63. Section B Membership data: This section seeks information on the membership of the total DB scheme, except for rows 195 to 206 which is restricted to the licensee. It is accepted that this data may need to be revised in subsequent years. If opening balances are revised, please correct the previous year's data rather than adjusting in the current year. Where any changes are made to prior year data, these should be detailed in the Commentary. Instructions for completion are as follows:

- Rows 139, 151, 158 & 167: These rows include values for members who became deceased at or during the financial year.
- Row 165: New dependents should be populated with any dependent members created on the death of a member in one of the other categories.
- Rows 180 -190: Enter membership totals split between those with protected rights under the Protected Persons legislation and the remainder of the scheme.
- Rows 195 to 206: Split the membership associated with the licensee between those with protected rights under the Protected Persons legislation and the remainder of the scheme. Provide explanation in the Commentary, including if any amounts have been accepted where assets do not cover the liabilities.

2.64. Section C Scheme mergers. Instructions for completion are as follows:

- Rows 215 244: This section provides details of any mergers into this DB scheme, collecting incoming scheme data to compare to scheme data pulled from other sections.
- Rows 246 247: The funding ratios refer to the percentage of scheme asset cover of scheme liabilities. This should be provided based on year-end values for assets and liabilities.

2.65. Section D Actuarial valuation results: This section seeks information on the actuarial valuation results for the total scheme. This refers to scheme funding valuations rather than valuations for other purposes (for example, company accounting valuations or PPF levy valuations). Where necessary answers should be expanded upon in the Commentary. Instructions for completion are as follows:

- Rows 268 269: Average remaining active service life should be provided for the scheme membership as a whole, however if there is a significant difference in the remaining service life between non-regulated and regulated members, please provide both numbers in the Commentary.
- Rows 282 283: Specify the mortality table used, including any age ratings: eg "PMA/PFA92 U=2004 x-1 with medium cohort improvement factors subject to a 1% underpin". Expand in Commentary as necessary.
- Rows 286 287: Specify the mortality table used, including any age ratings: eg "PMA/PFA92 U=2004 x-1" with medium cohort improvement factors subject to a 1% underpin. Expand in Commentary as necessary.
- Rows 296, 312, 328 & 344: The rows headed "normal retirement age" are to be completed with the earliest age at which unreduced benefits are payable on age retirement (which might strictly be different to the scheme"s "normal retirement age").
- Rows 357 361: Information on AVC facilities is required only to the extent that employees can purchase additional service in the defined benefit scheme or the employer contributes in some way. Information is not required for defined contribution AVC schemes.

2.66. AVC information is required for the scheme as a whole.

3. Instructions for completing the total expenditure worksheets

Chapter Summary

The purpose of this chapter is to provide information on the total expenditure worksheets by each TO. This is to enable Ofgem to effectively monitor the performance of the companies in relation to their business plans and the total expenditure baselines set in the Final Proposals.

Introduction

3.1. The purpose of the worksheets in this area is to report total expenditure. In the main these worksheets pull data from other areas of the RIGS.

Overview of worksheets

- 3.2. The worksheets included within this chapter are:
 - Summary Totex
 - Ofgem Adjustments to Totex

Worksheet – Summary Totex (No input required)

3.3. The purpose of this table is to collect a summary of total opex and capex data to allow the financial model to be populated with actual reported data for the year.

3.4. The table is self populated with information from the Ofgem adjustments worksheet.

Worksheet - Ofgem Adjustments to Totex (No input required)

3.5. The purpose of this table is to make any adjustments to the figures reported by transmission companies before they are fed into the financial model. These adjustments will be discussed with the licensees

3.6. The table is populated by Ofgem and licensees should not enter any data in here.

4. Instructions for completing the operating expenditure worksheets

Chapter Summary

The purpose of this chapter is to inform the completion of the operating expenditure worksheets by each TO. This is to enable Ofgem to effectively monitor the performance of the companies in relation to their business plans and operating expenditure baselines set in the Final Proposals.

Introduction

4.1. The purpose of the worksheets in this area is to report operating costs (opex) information at various different levels to enable Ofgem to fully understand opex trends and performance. Certain large and significant areas of cost are broken down into greater detail so that we can understand the movements more easily.

4.2. Licensees should submit accurate (and where instructed) audited figures of their costs and revenues for the relevant period. Further guidance is provided below.

4.3. All costs are to be entered on a cash basis (see Glossary) and exclusive of atypical items except where specifically instructed to report data. Cash means exclusive of all provisions and all accruals and prepayments that are not incurred as part of the ordinary level of business.

4.4. Forecast data will **only** be required at a total category level on the summary of cash controllable costs table.

Overview of worksheets

4.5. The worksheets included within this chapter are:

- Summary of cash controllable costs
- Provisions
- Irregular items
- Direct opex
- Analysis of excluded, consented, and de minimis services
- Analysis of CEO & other corporate costs

- Analysis of IS & telecoms costs
- IS & telecoms allocations
- Insurance costs
- Analysis of property costs by building
- Property cost allocations
- Analysis of business support costs
- Business support allocations
- Related party transactions
- Year on year movement in controllable costs
- Total transmission salary and FTE numbers
- Resilience
- Apprentices and training
- In sourced outsourced
- IRM expenditure
- NIA expenditure
- NIC expenditure
- Innovation revenue condition
- Innovation revenue input
- Physical security opex.

Worksheet – Summary cash controllable costs

4.6. The purpose of this table is to provide a breakdown of cash controllable costs into activities within business support, closely associated indirect and direct costs. The table also collects non controllable costs and provision movements to add to total accounting costs.

4.7. Cost data is to be input on a cash controllable cost basis. On table 1.3a data is to be input net of capitalisation of closely associated indirect costs and business support costs. (Direct costs are drawn directly from table 1.3b on a net basis, ie costs are allocated directly to direct opex and there is no further capitalisation). Table 1.3b data is to be input gross of capitalisation of closely associated indirect costs and businest support costs and business support costs.

4.8. SO costs should be reported separately from TO costs where appropriate.

4.9. Table 1.3c shows the difference between tables 1.3a and 1.3b. This table represents the business support and closely associated indirect costs capitalised. It is automatically populated.

Worksheet – Provisions

4.10. The purpose of this table is to collect details of the provisions that have affected the results so that Ofgem can understand any significant events happening in the year.

4.11. Data should be input as positive or negative values as appropriate. SO costs should be reported separately from TO costs where appropriate. Provisions are those defined under standard accounting terminology.

Worksheet – Irregular items

4.12. The purpose of this table is to identify irregular or one off costs over £0.5m included in the costs Table 1.3abc so that Ofgem can understand any significant events happening in the year. Irregular items are costs that are not part of the normal recurring business activities.

4.13. SO costs should be reported separately from TO costs where appropriate.

Worksheet – Direct opex

4.14. The purpose of this table is to show the amount of cash controllable operating costs spent on fault repairs, planned inspections and maintenance, vegetation management, operational property management, physical security, BT 21 CN teleprotection, offshore transmission projects and allowed innovation costs.

4.15. Data should be input into the cells that are highlighted in yellow. Costs will be total costs for each activity. The main items included within "other" should be identified in the cells highlighted in yellow.

Worksheet - Analysis of excluded, consented, and de minimis services

4.16. The purpose of this table is to collect costs and revenues relating to excluded, consented, and de minimis services provided by the transmission business by type of service.

Worksheet - Analysis of CEO & other corporate costs

4.17. The purpose of this table is to show the gross cash controllable cost of CEO and other corporate costs and the amounts allocated to GB regulated network businesses directly or via a related party and to other business within the group.

4.18. The costs should reflect the total costs relating to their GB regulated network businesses including excluded services associated with those businesses. The table should reflect the amount of costs allocated directly to the businesses with allocations of such functions, if any, from related parties.

Worksheet – Analysis of IS & telecoms costs

4.19. The purpose of this table is to collect gross cash controllable cost information relating to non-operational Information Systems (IS) & telecoms expenditure and activities.

4.20. Input costs for each IS & telecoms function by cost category. Input staff and contractor numbers in total.

Worksheet - IS & telecoms costs allocations

4.21. The purpose of this table is to show the allocation of costs to the GB regulated network businesses, excluded services and other non-regulated business, split between non-operational and operational costs (although it is expected the allocation will be 100% to non-operational).

4.22. There are no specific instructions for completion of the table.

Worksheet - Insurance costs

4.23. The purpose of this table is to collect cash controllable cost information relating to insurance costs, premiums, policies and cover etc, including the actual costs and cover relating to the transmission and other GB regulated network businesses.

4.24. In Table 2.7.1 Insurance Premiums - input the appropriate details of all insurance policies and how this is allocated to the businesses and excluded services. Insurance premiums - Cost of insurance premiums including insurance premium tax.

4.25. In Table 2.7.2 Total Insurance Department costs - input details of the total cost of the insurance department and how this is allocated to the businesses and excluded services.

Worksheet - Analysis of property costs by building

4.26. The purpose of the table is to collect gross cash controllable cost information relating to all non-operational property costs.

4.27. Buildings holding SO staff or assets should be treated as non-operational.

Worksheet - Property costs allocations

4.28. The purpose of this table is to show the allocation of costs to the GB regulated network businesses, excluded services and other non-regulated business, split between non-operational and operational costs.

4.29. There are no specific instructions for completion of the table.

Worksheet - Analysis of business support costs

4.30. The purpose of this table is to provide gross cash controllable cost analysis of business support costs and table 2.9.1 shows how the costs are charged to the GB regulated network businesses (and non regulated business where required). These tables show non-operational costs only.

4.31. Costs should be entered by business support cost type and type of expenditure in the boxes highlighted in yellow (several categories/lines are linked to other tables and updated automatically).

Worksheet - Business support allocations

4.32. The purpose of this table is to show the allocation of costs to the GB regulated network businesses, excluded services and other non-regulated business, split between non-operational and operational costs (although it is expected the allocation will be 100% to non-operational).

4.33. There are no specific instructions for completion of the table.

Worksheet - Related party transactions

4.34. The purpose of this table is to provide an analysis and understanding of the nature and size of services provided to the transmission business and other GB regulated network businesses by each related party.

4.35. Input a description of the services provided by each related party. Input as positive numbers the turnover data for the related party as charged to the transmission business, other regulated network businesses and external customers. Input as negative numbers the respective costs incurred. Where the total charge from a related party to the transmission business is less than £500k per annum that related party does not need to be included on this table.

Worksheet - Year on year movement in controllable costs

4.36. The purpose of this table is to enable licensees to explain the reasons for increases and decreases in costs year on year of $\pm 0.5m$ (these are the net increases and decreases year on year in table 1.3abc).

4.37. Licensees should fill in the reasons for changes in costs in the boxes shaded in yellow. The table should be completed to clearly explain the year on year movements additional explanations can be provided in the narrative if required.

Worksheet - Total transmission salary and FTE numbers

4.38. The purpose of this table is to show the total transmission and business support gross staff costs and FTEs. This will provide a cost per FTE for comparisons of total transmission employment costs and business support employment costs.

4.39. Licensee should fill in total gross staff costs by type, and FTEs as indicated by the boxes shaded in yellow. Trainee Craftsperson and Engineer FTE numbers and costs are linked to table 2.18 Apprentices and Trainees and automatically updated. Other Trainees represent Graduate and Other Staff/Management Trainees.

Worksheet – Resilience

4.40. The purpose of this table is to identify all opex and capex resilience costs covering all dimensions of resilience excluding physical and cyber security see table below.

4.41. Costs should be input as positive values. Where costs are specific to an asset or site the name of the site or asset should be given. All costs should relate to resilience costs incurred as stand-alone and/or retrofit expenditure whether on specific assets, specific sites or a collection of assets/sites. For the avoidance of doubts any resilience costs embedded within future schemes as part of enhanced standards and formal requirements (eg legal and environmental) need not be split out from relevant schemes and reported in these tables.

Worksheet – Apprentices and training

4.42. The purpose of this table is to show the numbers and costs of apprentices and other trainees and the training costs associated with them.

4.43. Costs should be input as positive values. Data should be input into the cells that are highlighted in yellow.

Worksheet – Insourced/Outsourced

4.44. The purpose of this table is to help in understanding the different business models in use by companies and how much work is done in house or by contractors. It will help in comparison of costs between TOs.

4.45. Enter details of internal labour, ongoing pension costs and directly procured materials. For all open book contracts enter the value of labour only contracts and other contracts breaking down the total contract cost into the various headings identified in the table. For all closed book contracts enter the total value of labour only contracts and the total value of other contracts.

Worksheet – Innovation Rollout Mechanism (IRM) expenditure

4.46. The purpose of this table is to provide Ofgem with the costs and other data relating to schemes designed to rollout a proven innovation.

4.47. There are two reporting tables within this worksheet. The first is for the scheme name and category, with total costs reported by year. The second is to split the total costs by cost type. These costs will be added to the RAV in accordance with the rules.

Worksheet – Network Innovation Allowance (NIA) expenditure

4.48. The purpose of this table is to report the amounts spent under the NIA. NIA expenditure information is required to monitor the total amount spent by the Licensee and claimed through the NIA funding mechanism.

4.49. Costs reported in this table must be incurred in accordance with the version of the NIA Governance Document in place when the expenditure was made, as published by Ofgem. In the description below, capitalised terms are defined in the NIA Governance Document. No expenditure should have been incurred (and therefore reported) prior to 1 April 2013. This worksheet requires the same data broken down by different categories. The first requires detailed cost data to be reported by cost type. The second breaks the costs down as either bid preparation Costs, or against each (named) project.

Worksheet – Network Innovation Competition (NIC) expenditure

4.50. The purpose of this table is to collect expenditure from the NIC project bank account for any NIC project that is being implemented. The expenditure is recorded by project. As NIC projects do not form any part of RIIO-T1 allowances and requires separate detailed reporting every 6 months, this table will serve only to balance this worksheet to the regulatory accounts.

4.51. This table should only show expenditure for the regulatory year and not the outstanding funds required for the project. No expenditure should have been incurred (and therefore reported) prior to April 2013.

Worksheet - Innovation revenue condition

4.52. The purpose of this table is to enable Ofgem to set the pass through rate for the NIA. No input is required.

Worksheet - Innovation revenue input

4.53. The purpose of this table is to inform Ofgem on licensees revenue associated with the innovation stimulus (NIC, NIA and IRM).

4.54. This table sets out the amount to be collected by the licensee for the regulatory year for the NIC. The amount will be the full amount for the whole duration of the Project and not just the expenditure in that regulatory year.

4.55. The table shows the amount collected by the licensee for the regulatory year for the NIA. The Licensee should report the amount that was recovered to fund NIA projects or prepare submissions for the NIC as part of their eligible bid preparation costs.

4.56. The table shows the amount to be collected by the licensee in each regulatory year through the IRM. The amount to be collected for the NIC and IRM is calculated in accordance with a Funding direction issued by the Authority in accordance with GDC 10.

Worksheet - Physical security

4.57. The purpose of this table is to inform Ofgem of the opex spent on physical security

4.58. There are no specific instructions for completion of the table.

5. Instructions for completing the capital expenditure worksheets

Chapter Summary

The purpose of this chapter is to inform the completion of the capital expenditure worksheets by each TO. This is to enable Ofgem to effectively monitor the performance of the companies in relation to their business plans and capital expenditure baselines set in the Final Proposals.

Introduction

5.1. The purpose of the worksheets in this area is to report capital expenditure (capex) information at various different levels to enable Ofgem to fully understand capex year on year.

5.2. Licensees should submit accurate and (where instructed) audited figures of their costs for the relevant period. Further guidance is provided below.

5.3. All costs are to be entered on a cash basis (see glossary) and exclusive of atypical items except where specifically instructed to report data. Cash means exclusive of all provisions and all accruals and prepayments that are not incurred as part of the ordinary level of business.

5.4. Forecast data will only be required on the capex summary worksheet.

Overview of worksheets

- 5.5. The worksheets included within this chapter are:
 - Capex summary
 - Non operational capex
 - System Operator (SO) capex
 - Load-related scheme listing
 - Load-related scheme annual profile
 - Non load-related scheme listing

- Non load-related scheme annual profile
- Other capex
- Flood mitigation
- TIRG / TII/ SWW schemes
- Unit costs actual
- Unit costs future levels
- Physical security capex.

Worksheet – Capex summary

5.6. The purpose of this table is to collect information relating to the licensee's historical and forecast expenditure.

5.7. There are no specific instructions for completion of the table.

Worksheet – Non operational capex

5.8. The purpose of this table is report expenditure on non operational capex.

5.9. The table requires the licensee to insert the name of specific IT projects where the total expenditure is £1m or more. Full project details, not just expenditure in the year, should be entered as indicated by the column headings. Expenditure on all other assets and IT assets less than £1m should be entered in total. Where the total spent on a project is more than £1m but the expenditure within a particular year is less than £1m, this should be shown as an individual project and not included in the IT expenditure <£1m category.

Worksheet – System Operator (SO) capex

5.10. The purpose of this table is to report the expenditure on SO capex.

5.11. Enter the name of each project. Data should be entered for each project under the appropriate asset heading.

Worksheet - Load-related scheme listing

5.12. The purpose of this table is to collect information relating to the load-related schemes within the capital expenditure programme of the licensee.

5.13. The definitions for all the categories are as defined in the glossary. The totals and subtotals in this table should all feed into table 4.18.1a capex summary. As a one-off exercise TOs are required to complete this table with actual outturn data for closed schemes over the last 5 regulatory years.

Worksheet - Load-related scheme annual profile

5.14. The purpose of this table is to collect information relating to the timing of loadrelated schemes within the capital expenditure programme of the licensee.

5.15. The information should be provided on a total cost basis. The relevant annual gross scheme cost including all allocated contingencies and overheads should be reported. The relevant annual average gross scheme cost for the period including all allocated contingencies and overheads should be reported.

Worksheet - Non load-related scheme listing

5.16. The purpose of this table is to collect information relating to the non loadrelated schemes within the capital expenditure programme of the licensee.

5.17. The definitions for all the categories will be defined in the glossary. The totals and subtotals in this table feed into table 4.18.1a capex summary.

Worksheet - Non load-related scheme annual profile

5.18. The purpose of this table is to collect information relating to the timing of non load-related asset replacement schemes within the capital expenditure programme of the licensee.

5.19. The information should be provided on a total cost basis. The relevant annual gross scheme cost including all allocated contingencies and overheads should be reported. The relevant annual average gross scheme cost for the period including all allocated contingencies and overheads should be reported.

Worksheet - Other capex

5.20. The purpose of this table is to collect information on TO capex schemes in the other categories of the capex summary table.

5.21. The definitions for all the categories will be defined in the glossary. The totals and subtotals in this table feed into table 4.18.1a capex summary.

Worksheet - Flood mitigation

5.22. The purpose of this table is to collect volumes and levels of expenditure on work to mitigate the risk of flooding.
5.23. TOs must input the number of assets in each risk category which have had, or are forecast to have, flood mitigation work carried out. This information provided will be shared with DECC.

Worksheet – Transmission Investment Renewable Generation (TIRG) / Transmission Investment Incentives (TII)/ Strategic Wider Works (SWW) schemes

5.24. The purpose of this table is to collect information relating to capital expenditure and completion dates of all TIRG / TII and SWW schemes.

5.25. There are no specific instructions for completion of the table.

Worksheet – Unit costs actual

5.26. The purpose of this table is to understand the make-up of the current non load-related capex spend between price and volume.

5.27. This table requires details of volumes and costs by asset type. The table seeks to understand average unit costs (\pounds m/annum) based on volumes and associated \pounds m unit costs of investment. This average unit cost can either be an input figure or a derived figure as appropriately reflects the licensee's costs. We require a clear explanation of how the licensee has determined the data within this table.

Worksheet - Unit costs future levels

5.28. The purpose of this table is to understand the make-up of the current non loadrelated capex spend and forecast for the price control period between price and volume.

5.29. This table requires details of volumes and costs by asset type. The table seeks to understand average unit costs (\pounds m/annum) based on proposed volumes and associated \pounds m unit costs of investment. This average unit cost can either be an input figure or a derived figure as appropriately reflects the licensee's estimation of costs. We require a clear explanation of how the licensee has determined the data within this table.

Worksheet - Physical security capex

5.30. The purpose of this table is to inform Ofgem of the capex spent on physical security.

5.31. There are no specific instructions for completion of the table.

6. Instructions for electricity network data worksheets

Chapter Summary

The purpose of this chapter is to inform the completion of the electricity network data worksheets by each TO. This is to enable Ofgem to effectively monitor the performance of the electricity transmission networks including in relation to system flows, boundary transfers and asset health during RIIO-T1.

Introduction

6.1. The purpose of the worksheets in this area is to report data on the network and its performance at various different levels to enable Ofgem to fully understand the network changes and network performance year on year.

6.2. Licensees should submit accurate and (where instructed) audited figures of their data for the relevant period. Further guidance is provided below.

Overview of worksheets

6.3. The worksheets included within this chapter are:

- System characteristics
- Activity indicators
- Activity costs
- Transmission system performance
- Faults
- Failures
- Boundary transfers and capability
- Boundary transfers and capability development
- Demand and supply at substations
- Asset description
- Asset age profile
- Asset disposals load-related
- Asset disposals non load-related
- Asset additions and disposals total (Aged Based Modelling)
- Asset additions and disposals total
- Asset replacement priority driven disposals

• Asset lives.

Worksheet - System characteristics

6.4. The purpose of this table is to collect high-level information relating to physical characteristics of the transmission network to understand both historic development and future development of the network and to provide a clear high-level view of the trajectory being proposed and justified relative to the network requirements and investment case.

6.5. All system characteristics should be entered as at the end of a reporting year (ie 31 March).

Worksheet – Activity indicators

6.6. The purpose of this table is to collect key indicators of the overall level of transmission activity. Please note that for demand and generation forecasts including reactive capacity each TO should provide their assumptions used in their network planning process consistent with the requirements of the National Electricity Transmission System Security and Quality of Supply Standards (NETS SQSS) for planning their network. If the assumptions differ from those provided by the GB system operator under the SO:TO Code (STC) a narrative should be provided explaining why different assumptions have been used.

6.7. There are no specific instructions for completion of the table.

Worksheet – Activity costs

6.8. The purpose of this table is to collect key indicators of the local activity as put forward by the TO in table 4.2.1a as local volume drivers; and to capture related investment.

6.9. There are no specific instructions for completion of the table.

Worksheet – Transmission system performance

6.10. The purpose of this table is to collect information relating to the quality of transmission service delivered.

6.11. There are no specific instructions for completion of the table.

Worksheet – Faults

6.12. The purpose of this table is to collect information about the faults on the transmission network.

6.13. There are no specific instructions for completion of the table.

Worksheet – Failures

6.14. The purpose of this table is to collect information about the failures on the transmission network.

6.15. For each asset category please specify the primary causes of failure and the number of failures associated with each cause.

Worksheet – Boundary transfers and capability

6.16. The purpose of this table is to collect information on transmission capacity against required transfer levels at key parts of the transmission system, as indicators of load-driven need for developing the transmission infrastructure.

6.17. There are no specific instructions for completion of the table.

Worksheet - Boundary transfers and capability development

6.18. The purpose of this table is to collect actual and forecast information on investments required for each boundary to deliver boundary capability deemed required on the basis of Table 4.8.2 and as identified within Table 4.19.1.

6.19. Costs provided within this table for wider works should comply with the following definition of 'Wider Works' as specified on Page 4 of the "Connect and Manage Guidance" document issued by National Grid in January 2011 'found at: http://www.nationalgrid.com/NR/rdonlyres/01463C70-F178-4930-9A00-780FE5330F2D/47332/CMversion50.pdf

Wider Works' are the other transmission reinforcement works (ie not Enabling Works) associated with reinforcing the network to accommodate the new generating station and ensure compliance with the NETS SQSS.

6.20. Boundaries: As defined in Table 4.8.1a above ie the same list of boundaries must appear in Table 4.8.2.

6.21. Boundary Capability (MW): Thermal, voltage and stability: capabilities for each boundary identified by the licensee assessed according to the NETS SQSS both before any investment and as delivered by investments identified in this table and table 4.8.2.

6.22. Dependencies: Please provide a short hand commentary on any key dependencies relating to scheme delivery of enhanced boundary capability eg completion of another scheme (provide scheme ID consistent with Table 4.19.1) or key generation or demand assumption(s) – please indicate. We will expect the narrative to expand on this.

Worksheet – Demand and supply at substations

6.23. The purpose of this table is to collect information relating to more localised demand which is driving need for capex.

6.24. The licensee should provide outturn data for the reporting year and historical years. Supply capacity is assessed in two different ways - as limited by Super Grid Transformer (SGT) capacity only, and as limited by any other factors. In both cases the assessment should be according to the NETS SQSS, for example, taking due account of any LV transfer capability and contribution from embedded generation.

Worksheet – Asset description

6.25. The purpose of this table is to collect information relating to the licensee's asset population. This is largely used to analyse the non load-related expenditure of the licensee. This is an information table only, which will be used to provide clarity on the asset categories referred to in other tables.

6.26. The list of assets is indicative only and data should be provided in accordance with the TO's own system for classifying and grouping assets for asset life based planning purposes. However, there should be reasonable stability in data types over the years to facilitate assessment of trends etc. The licensee should give an explanation in the accompanying narrative for any changes in asset categories that occurred in the reporting year and provide information on mapping the historical classification to any new classification.

Worksheet – Asset age profile

6.27. The purpose of this table is to collect information relating to the licensee's asset population age profiles for asset replacement modelling purposes.

6.28. Input information as indicated by the yellow shaded boxes on the table.

Worksheet – Asset disposals load-related

6.29. The purpose of this table is to collect information relating to the disposal of assets on the licensee's network included in load-related schemes. This table is used to assess the Network Output Measures (NOMs).

6.30. Input information as indicated by the yellow shaded boxes on the table.

Worksheet – Asset disposals non load-related

6.31. The purpose of this table is to collect information relating to the disposal of assets on the licensee's network included in non load-related schemes. This table is used to assess the NOMs.

6.32. There are no specific instructions for completion of this table.

Worksheet - Asset additions and disposals total (Aged Based Modelling)

6.33. The purpose of this table is to enable Ofgem to populate its aged based model and understand the variances year on year and from the business plan.

6.34. There are no specific instructions for completion of this table

Worksheet – Asset additions and disposals total

6.35. The purpose of this table is to collect information relating to the disposal of assets on the licensee's network included in load-related schemes. This table is used to assess the NOMs.

6.36. The list of assets is indicative only and data should be provided consistent with the asset categories as defined in Table 4.27.1 and 4.27.3.

Worksheet – Asset replacement priority driven disposals

6.37. The purpose of this table is to identify all asset replacement related disposals initially identified by the TO based on assessment of Replacement Priority (RP) categorisation before consideration of wider practical factors – such as outages, resources, system drivers and scheme bundling, which lead to the best asset replacement plan identified in Table 4.15.1.

6.38. For the avoidance of doubt the data in this table should be consistent with that which would be expected to be derived following completion of the asset
`Replacement Priorities" stage (capturing `network risk') as illustrated under Figure 1 of the formal TO Joint Methodology document found at: <a href="http://www.ofgem.gov.uk/Networks/Trans/RegReporting/Documents1/Proposed%20Network%20Netw

Worksheet – Asset lives

6.39. The purpose of this table is to collect information relating to the licensee's asset population lives for asset replacement modelling purposes.

6.40. Asset lives should be based on the known and planned levels of refurbishment which results in an increase in the average expected remaining life. The licensee should provide an explanation and any justifying evidence in the accompanying narrative for any changes in its estimate of the asset lives. The list of assets is indicative only and data should be provided consistent with the asset categories as defined in Table 4.12 Asset Age Profile.

7. Instructions for completing the outputs worksheets

Chapter Summary

The purpose of this chapter is to inform the completion of the outputs worksheets by each TO. This is to enable Ofgem to effectively monitor the performance of the companies in delivering their RIIO-T1 outputs and to determine any associated reward or penalty under the incentive arrangements consistent with the Final Proposals.

Introduction

7.1. The purpose of the worksheets in this area is to report performance against the various outputs. This will allow Ofgem to monitor licensee's performance against the output targets year on year.

7.2. Licensees should submit accurate and (where instructed) audited figures of their data for the relevant period. Further guidance is provided below.

7.3. All costs are to be entered on a cash basis (see Glossary) and exclusive of atypical items except where specifically instructed to report data. Cash means exclusive of all provisions and all accruals and prepayments that are not incurred as part of the ordinary level of business.

7.4. Some of the existing electricity transmission licence conditions that are subject to change at RIIO-T1 have existing reporting requirements that overlap with some of the proposed output worksheets. We have not as yet reviewed the degree to which some of the existing reporting requirements may still be needed. We will do this before the publication of the final RIGs.

Overview of worksheets

7.5. The worksheets included within this chapter are:

- Safety
- Reliability
- Customer satisfaction
- Timely connections
- Environment SF₆ emissions
- Environment visual amenity designated areas
- Strategic Wider Works (SWW) outputs

- Pre-construction deliverables NGET
- Environment Business Carbon Footprint
- Wider works volume driver
- NGET local generation and demand volume driver summary
- NGET local generation volume driver scheme level detail
- NGET planning requirements volume drivers (Uncertainty Mechanisms)
- NGET DNO mitigation schemes scheme level detail
- Local demand volume driver scheme level detail
- NGET DNO Mitigations schemes scheme level detail
- NGET undergrounding volume driver scheme level detail
- SHETL local generation volume driver
- Revenue driver information
- Collector substations volume driver
- Network Output Measures (NOMs)
- NOMS NGET
- Criticality substations
- Criticality circuits
- Criticality SPTL

Worksheet – Safety

7.6. The purpose of this table is to collect information on the costs each TO incurs in delivering its safety obligations.

7.7. In line with our Strategy Decision we have not stipulated an exhaustive list of legislative requirements but rather included examples of those obligations. This is to ensure this output remains relevant should any further legislative requirements be imposed on the businesses during the price control period.

7.8. To complete the worksheet each TO is required to:

- identify the costs associated with delivering the specific safety obligation identified
- if appropriate add other categories or safety obligations and the associated costs of delivering those obligations
- provide its total costs for meeting its safety obligations in each relevant year.

Worksheet – Reliability

7.9. The purpose of this table is to collect information in relation to incidents on the licensee's transmission system and the volume of unsupplied energy that is a consequence of these interruptions for the calculation of the licensee's Energy Not Supplied (ENS) incentive.

7.10. To complete the worksheet each TO is required to give details of:

- the total number of transmission system incidents that occurred during the year, the number of the incidents with a duration of more than three minutes, and the number of incidents categorised as excluded events
- the volume of energy that was not supplied to customers as a result of the total number of incidents, the volume of energy not supplied for incidents with a duration of more than three minutes, and the volume of ENS for incidents categorised as excluded events.

Worksheet – Customer satisfaction

7.11. The purpose of this table is to collect the results from surveys that the transmission owners are required to carry out under the customer/stakeholder satisfaction output. The output has two financial incentive elements. One is a stakeholder engagement discretionary reward. The second is a deterministic financial incentive rewarding or penalising the transmission owners for performance as appraised by its customers and/or stakeholders through survey.¹ This table relates to this second element reflecting survey results and supporting information. At this stage the supporting information is work in progress. This is information that will provide us with confidence that the overall reward through this financial incentive is appropriate.

7.12. To complete the worksheet each TO is required to include:

- baseline level of performance (as specified in Special Licence Condition ETC 24)
- mean, median and 25% and 75% quartile survey scores² (in the case of National Grid this includes both the survey of customers and the equivalent scores for the survey of stakeholders).

7.13. In December 2012, we will also include details of supporting information that each TO will be required to report on. This might differ for each TO.

¹ NGET and NGGT shall include results from both a survey of customers and a survey of stakeholders. SPTL and SHETL shall survey stakeholders.

² The median and quartile scores will give us a clearer view of the spread of responses to the survey.

Worksheet – Timely connections

7.14. The purpose of this table is to summarise the performance of the TOs in delivering timely connections to the network. The data will support the timely connections output based on delivery against licence obligations (and for SPTL and SHETL obligations in the SO:TO Code (STC)).

7.15. Each TO will have to identify the total number of connections by category of generation type where it made an offer. For NGET, this means the obligation contained in condition C8 paragraph 3. For SPTL and SHETL, this means the obligation contained in D4A of their licence conditions. It will need to report whether delivery against required timings were met.

7.16. To enrich our information, and in part to inform the work to develop RIIO-T2 incentives in this area, TOs will also need to report on whether further amendment was made following the delivery of the connection offer (or equivalent stage for SPTL and SHETL). Such amendment is not necessarily a sign of a problem. A commentary will be sought to provide the reason for the change.

Worksheet – Environment SF₆ emissions

7.17. The purpose of this table is to collect information in relation to emissions of sulphur hexafluoride from assets forming the licensee's transmission system for the annual calculation of the licensee's SF₆ incentive.

7.18. To complete the worksheet each TO is required to give details of:

- the total amount of SF_6 gas (in kilograms) contained in assets forming the licensee's transmission system at the start of the year
- the additions and disposals of SF_6 gas (in kilograms) in assets forming the licensees transmission system during the year
- the total emissions of SF₆ gas over the year the licensee has measured from its transmission system in accordance with the licensee's methodology statement
- the amount of any SF_6 emissions that the Authority has directed is the result of an exceptional event.

Worksheet – Visual amenity designated areas

7.19. The purpose of this table is to collect information on the delivery of mitigation measures to improve the visual amenity of Areas of Outstanding Natural Beauty, National Parks and National Scenic Areas scheme. The information worksheet will collect detail on mitigation activity volumes and the expenditure incurred.

7.20. The TO is to set out the length of overhead lines within each of the Designated Areas within their network.

7.21. Where no activity has been undertaken in a particular Designated Area, these cells should remain blank.

Worksheet – Strategic Wider Works (SWW) outputs

7.22. The purpose of this table is to collect information on the licensee's construction progress and completion of Strategic Wider Works that have been approved for funding by the Authority. The information worksheet will collect information on progress against construction milestones, annual expenditure profile, delivered outputs and performance against timely delivery standards.

Worksheet – Pre-construction deliverables

7.23. The purpose of this table is to collection information on the licensee's progress and expenditure on pre-construction engineering works. The information worksheet will collect information on the status of various pre-engineering activities in relation to a potential Strategic Wider Works scheme.

Worksheet – Environment Business Carbon Footprint

7.24. The purpose of this table is to provide information on the licensee's Business Carbon Footprint (BCF) (in tonnes of CO_2 equivalent) in order for us to review the carbon footprint across all the TOs, and to assess the year on year percentage change.

 $7.25. \text{CO}_2$ equivalent emissions arising from losses on the licensee's transmission system are included in the table since to provide an annual estimate of total BCF that will align with other environmental reports produced by the company.

7.26. The associated commentary must contain the methodology used, including detailed emission tables for each of the sections below, and further information on the methodology adopted.

7.27. As stated below, a licensee can select the 12 month period that it intends to use as its confirmed reporting year. This confirmed reporting year must have prior approval from Ofgem. We expect the confirmed reporting year to align with the statutory or regulatory accounts.

General principles of the reporting methodology

7.28. The reporting methodology must be compliant with the principles of the Greenhouse Gas Protocol³ ("GHG Protocol"). In summary⁴, the BCF reporting must be:

- Relevant: the inventory must reflect the substance and economic reality of the company's business relationships, not merely its legal form.
- Complete: all relevant emission sources must be included.
- Consistent: accounting approaches, inventory boundary and calculation methodology must be applied consistently over time.
- Transparent: information on the processes, procedures, assumptions and limitations of the BCF reporting must be disclosed in a clear, factual, neutral and understandable manner, enabling internal and external verifiers to attest to its credibility.
- Accurate: GHG measurements, estimates, or calculations must be systemically neither over nor under the actual emissions value, as far as can be judged, and that uncertainties be reduced as far as practicable.

Reporting boundaries

7.29. Licensee's must report on all Scope 1 and Scope 2 emissions (and a subset of Scope 3 emissions, as detailed below) on an "operational control" basis, ie report all emissions from operations on which the TO has full authority to introduce and implement its operating policy.

7.30. Licensees must also report on a subset of Scope 3 emissions (business travel and external contractors), to ensure that the reporting captures all of the emissions arising from the development and operation of the licensee's Transmission System, regardless of the legal entity carrying out each activity. According to this, we consider it valuable to focus on contractors emissions relating to the operational transport fleet and mobile power plants.

7.31. A licensee that forms part of a larger corporate group must provide a brief introduction outlining the structure of the group. The commentary must detail which organisations are considered to be within the reporting boundary for the purpose of this exercise.

7.32. Apportionment of emissions across a corporate group to the licensee's business units must be made clear in the commentary.

Contractors

7.33. The exclusion of any contractors must be justified and any thresholds used for exclusion must be stated in the commentary.

7.34. The commentary must also include an indication of what proportion of contractors have been excluded. This figure could be calculated based on contract value.

7.35. As far as possible, the licensee must try to ensure that data provided from different contractors is based on consistent assumptions.

Detailed reporting requirements

7.36. Licensees are given flexibility to set their own standards for:

- Reporting year we expect this generally to align with the statutory or regulatory accounts.
- The use of estimates rather than direct measurement, and any exclusion from the reporting based on (lack of) materiality considerations. Any assumptions used to make estimates must be included in the commentary. It is anticipated that data will need to be estimated under two scenarios:
 - when the type of emissions is not measured
 - when there is measurement data, but an estimate is required as the data is not at the same level of granularity as required by the summary BCF worksheet.

7.37. As a general principle, the licensees must focus more on the first type of estimation.

7.38. The commentary must include data tables for each area of emissions (ideally at the same level of granularity as the Defra conversion factors) containing the following information:

- the licensee in question
- the level of emissions (in tCO2e)
- the data source and collection process
- the relevant physical units eg miles
- the emission conversion factor used
- the source of the emission conversion factor (this shall be Defra unless there is a compelling case for using another conversion factor)
- the Scope of the emissions ie Scope 1, 2 or 3
- whether the emissions have been measured or estimated
- any tools used in the calculation
- whether the emissions stem from contractors.

7.39. The commentary must also include details of any auditing a licensee has performed to verify their emissions data.

Apportionment

7.40. When the emissions data is not available for the individual licensee then the apportionment factor used must be transparent. The basis for calculating the apportionment factor must also be included in the commentary.

7.41. We expect that the basis for calculating the apportionment factor will vary according to the area of emissions. The table below gives the preferred basis for determining the apportionment factor. Other methodologies can be used, but they must be justified.

Table 7.1: Apportionment factor determination		
Area of emissions	Basis for apportionment factor	
Building usage	Head count	
Operational Transport	Network length or km ² of the DSA.	
Business transport	Head count, or like operational transport	
Substations usage	Number of substations	
	Area of emissions Building usage Operational Transport Business transport	

Buildings energy usage

7.42. Emissions for electricity usage in buildings must be converted according to the factor for the "Grid Rolling Average".

7.43. Natural Gas, Diesel and other fuels are all categorised as fuel combustion and must be converted to tCO2e on either a Gross Calorific Value (Gross CV) or Net Calorific Value (Net CV) basis. We expect that this element of the chosen approach is clearly stated in the commentary and that this is consistently applied over time.

7.44. Electricity usage in substations must be captured under "Buildings energy usage". All substation consumption must be treated as energy supplied rather than losses. It is recognised that not all substations will be metered; rather, it is expected that licensees will in time register all substation as unmetered supplies and develop a common method for estimating consumption. Each licensee must include in its methodology (included in the commentary) the basis on which energy supplied has been assessed. Estimation could be based on a bottom-up approach, whereby the substation energy usage is split into estimates of its constituent parts, such as heating and lighting etc.

Transport

7.45. Defra guidelines provide for a range of emission conversion factors for transport means, with the aim to provide the best possible estimate of emissions from the vehicle portfolio owned and/or operated by the company. The reporting must, as far as reasonably practicable, use the full range of emission conversion factors available (as applicable to the range of means of transport actually used by the company).

7.46. Defra allows for transport to be entered in terms of both mileage and fuel consumption. Reporting must be based upon mileage, using conversion factors at the greatest level of disaggregation that is reasonably practicable. Reporting can be based on fuel consumption only where detailed and reliable data is available, eg through fuel cards.

7.47. In cases where emission factors for specific transport means are not available (we are aware of this issue for helicopters, but there may be some other instances) the equivalent tonnes of carbon dioxide (tCO2e) must be estimated and summed to the closest means of transport (eg "air" for helicopters). The methodology and assumptions used for estimating/measuring these emissions must be included in the commentary.

7.48. Operational Transport is the transportation (often a fleet of vehicles) used in the day to day operation of the business – ie in the inspection and maintenance of the network.

7.49. Business Transport is that undertaken by staff travelling to locations that are other than their normal place of work or moving between sites for purposes such as meetings.

Fugitive emissions

7.50. This category caters for GHG emissions from a range of gases that may be relevant to the TO business. We anticipate that this will mainly include SF_6 emissions, but other gases may be included (eg HFC from air conditioning). SF_6 emissions must be reported in accordance with ENA-ER S38, using Defra conversion factors.

7.51. The commentary must identify which fugitive emissions have not been calculated or estimated.

Fuel combustion (non-building)

7.52. This is to cover for non-building fuel usage, such as mobile plants and the stand-by diesel mobile generators that are deployed from time to time in response to planned outages or faults. Defra emissions factors must be used. All mobile plant and generation used by the licensee, related and affiliate undertakings, contactors and sub-contractors must be included in so far as it is reasonably practicable. The methodology must describe the degree of estimation, and decisions to exclude any sources of emissions, applied.

Losses

7.53. This is to consider TOs responsibility towards losses as a Scope 2 emission, using the Defra conversion factor "Grid Rolling Average" for electricity losses. Substation electricity usage must be excluded from the reported emissions for network losses, so that it is not double-counted.

7.54. As explained above, an estimate of losses is required because it is important for Ofgem to have an indication of the overall BCF for a company and across all TOs.

Worksheet - NGET wider works volume driver

7.55. The purpose of this table is to collect information on schemes that are subject to the wider works volume driver. This information will allow us to adjust the TOs' baseline to reflect the actual capability delivered.

7.56. This table should only include details of schemes that are in the Network Development Policy (NDP) process. For the avoidance of doubt this excludes Strategic Wider Works (SWW) schemes. Information should be provided for both schemes that have already been delivered during the period and those that are currently within the construction phase (work in progress). A short commentary should be provided for any changes that have occurred during the reporting year detailing the reasons and impact of these changes. You should also confirm whether all schemes are still compliant with the NDP process.

7.57. Boundary Capacity (MW): This should be the capability for each boundary identified by the licensee assessed according to the NETS SQSS. The boundary capacity reported should be the limiting capacity for each of the boundaries. We only require the actual boundary capacity at start of year to be reported in the year 2013-14. Subsequent years will be calculated automatically. The actual capacity for each scheme should be detailed in the year the output has either been delivered or expected to be delivered.

7.58. Costs: The cost of the schemes should not include any preconstruction costs associated with these schemes as this proportion of the costs is not subject to the volume driver. The total scheme costs are made up of the annual profile of scheme costs incurred during RIIO-T1, the proportion of costs incurred pre-RIIO-T1 and the forecast of any costs that will be incurred in RIIO-T2.

Worksheet – NGET local generation and demand volume driver - summary

7.59. The purpose of this table is to collect information on the actual outputs and expenditure associated with the demand related infrastructure (shared use) and local generation (shared use) volume drivers. This information will allow us to adjust the licensee's baseline to reflect the actual outputs delivered.

7.60. Information should be provided for schemes that have been complete and schemes are currently in the process of being delivered.

7.61. Outputs: The table requires a summary of total outputs delivered for each output category. For the avoidance of doubt outputs should only be considered delivered once the overall scheme has been completed (commissioned). The tables include a number of categories for underground cables based on discrete lengths. The following criteria should be used in filling out this part of the table.

• the 3km length applies to all routes less than 3km

- the 15km length will apply to all routes between 3km and 15km
- the 75km length will apply to all other routes.

Worksheet – NGET local generation volume driver – scheme level detail

7.62. The purpose of this table is to collect scheme specific cost and output information on those schemes subject to the local generation volume driver.

7.63. This table is split into two sections, schemes that have been delivered up to the current reporting year and those schemes that are currently being constructed.

7.64. Scheme name: The scheme name and references should be detailed in such a way that will allow us to cross reference information in Table 4.19.1.

7.65. Year of delivery: The year of delivery should be the year that the scheme has been commissioned.

7.66. Costs: The table should include both the actual costs delivered up to the current reporting year and a forecast of costs that are anticipated in future years. The total scheme costs are made up of the annual profile of scheme costs incurred during RIIO-T1, the proportion of costs incurred pre-RIIO-T1 and the forecast of any costs that will be incurred in RIIO-T2.

7.67. Outputs: The table includes the outputs that make up the local generation volume driver. This information is detailed at a scheme specific level. The table includes a number of categories for underground cables based on discrete lengths. The following criteria should be used in filling out this part of the table.

- the 3km length applies to all routes less than 3km
- the 15km length will apply to all routes between 3km and 15km
- the 75km length will apply to all other routes.

Worksheet – NGET planning requirements volume drivers (Uncertainty Mechanism)

7.68. The purpose of this table is to collect information on the actual outputs and expenditure associated the planning requirements volume drivers.

7.69. This information will allow us to adjust your baseline to reflect the actual outputs delivered. Information should be provided for schemes that have already been delivered during the period and forecasts of schemes that will be delivered in the future.

7.70. Outputs: The table requires a summary of total outputs delivered for each output category. For the avoidance of doubt outputs should only be considered delivered once the overall scheme has been completed (commissioned) and not when individual outputs within a scheme are delivered.

7.71. The tables include a number of categories for underground cables based on discrete lengths. The following criteria should be used in filling out this part of the table:

- the 3km length applies to all routes less than 3km
- the 15km length will apply to all routes between 3km and 15km
- the 75km length will apply to all other routes.

Worksheet – NGET DNO mitigation schemes - scheme level detail

7.72. The purpose of this table is to collect scheme specific cost and output information on those schemes subject to the DNO mitigation schemes volume driver.

7.73. This table is split into two sections, schemes that have been completed and those schemes that are currently in the process of being delivered.

7.74. Scheme name: The scheme name and references should be detailed in such a way that will allow us to cross reference information in Table 4.19.1.

7.75. Year of delivery: The year of delivery should be the year that the scheme has been commissioned.

7.76. Costs: The table should include both the actual costs delivered up to the current reporting year and a forecast of costs that are anticipated in future years that are needed to deliver the scheme. The total scheme costs are made up of the annual profile of scheme costs incurred during RIIO-T1, the proportion of costs incurred pre-RIIO-T1 and forecast of any costs that will be incurred in RIIO-T2.

7.77. Outputs: The table includes the outputs that make up the DNO mitigation schemes volume driver. This information is detailed at a scheme specific level.

Worksheet – NGET undergrounding volume driver – scheme level detail

7.78. The purpose of this table is to collect scheme specific cost and output information on those schemes subject to the undergrounding volume driver.

7.79. This table is split into two sections, schemes that have been completed and those schemes that are currently in the process of being delivered.

7.80. Scheme name: The scheme name and references should be detailed in such a way that will allow us to cross reference information in Table 4.19.1.

7.81. Year of delivery: The year of delivery should be the year that the scheme has been commissioned.

7.82. Costs: The table should include both the actual costs delivered up to the current reporting year and a forecast of costs that are anticipated in future years that are needed to deliver the scheme. The total scheme costs are made up of the annual profile of the undergrounding costs incurred during RIIO-T1, the proportion of costs incurred pre-RIIO-T1 and forecast of any costs that will be incurred in RIIO-T2.

7.83. Outputs: The table includes the outputs that make up the undergrounding volume driver. This information is detailed at a scheme specific level.

7.84. The tables include a number of categories for underground cables based on discrete lengths. The following criteria should be used in filling out this part of the table:

- the 3km length applies to all routes less than 3km
- the 15km length will apply to all routes between 3km and 15km
- the 75km length will apply to all other routes.

Worksheet – SHETL local generation volume driver

7.85. The purpose of this table is to collect information on schemes that are subject to SHETL's connection volume driver. This information will allow us to adjust your baseline to reflect the actual capability delivered.

7.86. This table is split between shared use connections infrastructure and sole-use connections infrastructure with sub categories for typical and a-typical schemes. Information should be provided for both schemes that have already been delivered during the period and forecasts for schemes that will be delivered in the future.

7.87. Scheme name: The scheme name and references should be detailed in such a way that will allow us to cross reference information in Table 4.19.1.

7.88. Year of delivery: The year of delivery should be the year that the scheme has been commissioned.

7.89. Generation Connections: the amount of sole-use and shared-use capacity associated with generation delivered by individual schemes. This output should be recorded in the reporting year the scheme is expected to be complete.

7.90. Scheme Cost: The table should include both the actual costs delivered up to the current reporting year and a forecast of costs that are anticipated in future years for each scheme. The total scheme costs are made up of the annual profile of costs incurred during RIIO-T1, the proportion of costs in pre-RIIO-T1 and forecast of any costs that will be incurred in RIIO-T2.

Worksheet – Generation connections SPTL

7.91. The purpose of this table is to collect information on the actual volume and cost of new generation connections compared with forecasts for RIIO-T1.

7.92. The actual length of overhead line lengths and undergrounding is required for each scheme listed in the table. The table is split by sole-use connections and shared-use connections.

7.93. Please complete the table with relevant scheme information and highlight in your narrative any substitution of schemes since the submission of your RIIO-T1 business plan.

Worksheet – Collector substations volume driver

7.94. The purpose of this table is to collect information on the actual costs of new collector substations and associated overhead line lengths and platform civil costs.

7.95. Please complete the table with relevant scheme information.

Worksheet - Network Output Measures (NOMS)

7.96. The purpose of this table is to collect secondary deliverable information on current asset health and replacement priorities to assess performance against business plans and reported expenditure.

7.97. Input information as indicated by the yellow shaded boxes on the table.

Worksheet – NOMS NGET

7.98. The purpose of this table is to collect NGET secondary deliverable information on current asset health and replacement priorities to assess performance against business plans and reported expenditure.

7.99. Input information as indicated by the yellow shaded boxes on the table.

Worksheet – Criticality substations

7.100. The purpose of this table is to collect criticality information on substation assets as part of network output measures.

7.101. Input information as indicated by the yellow shaded boxes on the table.

Worksheet – Criticality circuits

7.102. The purpose of this table is to collect criticality information on circuits as part of network output measures.

7.103. Input information as indicated by the yellow shaded boxes on the table.

Worksheet – Criticality SPTL

7.104. The purpose of this table is to collect system criticality information as part of network output measures.

7.105. Input information as indicated by the yellow shaded boxes on the table.

8. Instructions for completing the revenue reporting worksheets

Chapter Summary

The purpose of this chapter is to inform the completion of the revenue reporting worksheets by each TO. This is to enable Ofgem to determine allowed revenues for the TOs.

Introduction

Whilst we have performed some initial development, the revenue reporting pack will be developed fully once the licence conditions are finalised. We anticipate providing a draft for comment in January of 2013.

Appendices

Index

Appendix	Name of Appendix	Page Number
1	Glossary	55

Appendix 1 – Glossary and Definitions

The purpose of this appendix is to provide definition of the terms included in these instructions and in the associated worksheets.

The definitions will be based upon those in the business plan data temples and will be completed jointly by Ofgem and licensees in January 2013 as part of finalising the RIGs.